Author’s response to reviews

Title: Effects of pre-notification, invitation length, questionnaire length and reminder on participation rate: a quasi-randomised controlled trial

Authors:

Marie Koitsalu (marie.koitsalu@ki.se)
Martin Eklund (martin.eklund@ki.se)
Jan Adolfsson (jan.adolfsson@ki.se)
Henrik Grönberg (henrik.gronberg@ki.se)
Yvonne Brandberg (yvonne.brandberg@ki.se)

Version: 3 Date: 04 Dec 2017

Author’s response to reviews:

Dear Faye Nourollahi,

Thank you for the opportunity to review our manuscript, "Effects of pre-notification, invitation length, questionnaire length and reminder on participation rate: a quasi-randomised controlled trial". As suggested, we have carried out the essential revisions. The manuscript has undergone a minor revision.

Responses to suggestions from reviewer/editor:

1. Please revise your Ethics statement to confirm all study participants provided informed consent to participate in your study.

   - The sentence "All participants provided written informed consent" has been added to the Ethics statement.

With kind regards,

Marie Koitsalu